Author's response to reviews

Title: Impact of urinary incontinence on health-related quality of life, daily activities, and healthcare resource utilization in patients with neurogenic detrusor overactivity.

Authors:

Derek H Tang (dtang@pharmacy.arizona.edu)
Danielle Colayco (dcolayco@komotohealthcare.com)
James Piercy (James.Piercy@adelphigroup.com)
Vaishali Patel (Patel_Vaishali@allergan.com)
Denise Globe (Globe_Denise@allergan.com)
Michael B Chancellor (chancellormb@gmail.com)

Version: 2 Date: 14 November 2013

Author's response to reviews:

Dear Mr Josefino Rodis et al.:

Thank you very much for these reviewer comments. Overall we found the comments to be quite helpful and they have further strengthened the manuscript. Detailed in the response to reviewer letter are our responses to these comments and how we have edited the manuscript. Cited references for specific responses are provided immediately under the corresponding paragraph. Please let me know if you have any questions. Thank you.

Sincerely,
Derek Tang